A Repurposing Strategy for Hsp90 Inhibitors Demonstrates Their Potency against Filarial Nematodes

Victoria Gillan¹, Kerry O'Neill¹, Kirsty Maitland¹, Francis M. Sverdrup², Eileen Devaney¹*

1 Institute of Biodiversity, Animal Health and Comparative Medicine, University of Glasgow, Glasgow, United Kingdom, 2 Center for World Health & Medicine, Saint Louis University, Saint Louis, Missouri, United States of America

Abstract

Novel drugs are required for the elimination of infections caused by filarial worms, as most commonly used drugs largely target the microfilariae or first stage larvae of these infections. Previous studies, conducted *in vitro*, have shown that inhibition of Hsp90 kills adult *Brugia pahangi*. As numerous small molecule inhibitors of Hsp90 have been developed for use in cancer chemotherapy, we tested the activity of several novel Hsp90 inhibitors in a fluorescence polarization assay and against microfilariae and adult worms of *Brugia in vitro*. The results from all three assays correlated reasonably well and one particular compound, NVP-AUY922, was shown to be particularly active, inhibiting Mf output from female worms at concentrations as low as 5.0 nanomolar after 6 days exposure to drug. NVP-AUY922 was also active on adult worms after a short 24 h exposure to drug. Based on these *in vitro* data, NVP-AUY922 was tested *in vivo* in a mouse model and was shown to significantly reduce the recovery of both adult worms and microfilariae. These studies provide proof of principle that the repurposing of currently available Hsp90 inhibitors may have potential for the development of novel agents with macrofilaricidal properties.

Citation: Gillan V, O'Neill K, Maitland K, Sverdrup FM, Devaney E (2014) A Repurposing Strategy for Hsp90 Inhibitors Demonstrates Their Potency against Filarial Nematodes. PLoS Negl Trop Dis 8(2): e2699. doi:10.1371/journal.pntd.0002699

Editor: Charles D. Mackenzie, Michigan State University, United States of America

Received September 11, 2013; Accepted January 2, 2014; Published February 13, 2014

Copyright: © 2014 Gillan et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: Work at the University of Glasgow was supported by grants from the Wellcome Trust (076734/Z/05/Z), wellcome.ac.uk, and the BBSRC (BB/E013473/1), bbsrc.ac.uk. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: eileen.devaney@glasgow.ac.uk

Introduction

Infections caused by the parasitic filarial nematodes Wuchereria bancrofti, Brugia malayi and Onchocerca volvulus remain a significant cause of pathology in the tropics. The adult stages of these pathogens are extremely difficult to kill with currently available drugs. Treatment relies upon two compounds, ivermectin (IVM) or diethylcarbamazine (DEC), both of which largely target the larval stage of the life cycle (the microfilariae, Mf). In the Global Campaign for the Elimination of Lymphatic Filariasis, either DEC or IVM is combined with albendazole. While this approach effectively disrupts transmission [1], Mf repopulate the circulation, necessitating the repeated administration of drug. As the reproductive life span of the adult female worm is estimated to be around 10 years for the lymphatic species [2] and longer for Onchocerca volvulus [3], programs aimed at eradication of these parasites are faced with a considerable challenge, as treatment must be continued over this long timescale. At least for O. volvulus, the repeated administration of ivermectin over many years is associated with treatment failures [4], although whether these truly reflect resistance remains the subject of debate. Consequently, drugs that target adult filarial worms would be a major advantage in control programs aimed at eliminating these parasites [5].

Heat shock protein 90 (Hsp90) has emerged in recent years as a validated target for the therapy of various tumors [6], resulting in the development of many Hsp90-specific small molecule inhibitors. Hsp90 is essential in all eukaryotes and several recent studies have demonstrated the activity of specific inhibitors against a

variety of tropical pathogens, such as *Plasmodium* [7,8], *Trypanosoma* sp [9] *Leishmania* sp [10] and the filarial worm *Brugia* [11,12]. The repurposing of compounds designed for one purpose to control of tropical infections is an attractive proposition [13], generating considerable enthusiasm in the pharmaceutical industry. Starting the search for new therapeutics for these diseases with drug-like molecules offers several short cuts, as these have already passed the basic criteria for development, have usually been optimized for their drug-like qualities and have often undergone toxicity testing.

Here we compare the efficacy of several classes of Hsp90 inhibitor against the lymphatic filarial nematode Brugia. The prototype Hsp90 inhibitor is geldanamycin (GA), a fermentation product of Streptomyces species that binds at the N-terminal ATP domain of Hsp90 disrupting its function [14]. Hsp90 acts as a molecular chaperone helping to fold and stabilize a variety of different proteins, the so-called 'client' proteins, many of which are involved in signal transduction [6]. The realization that Hsp90 client proteins, such as those encoded by oncogenes, were unable to attain their active conformation and were degraded following exposure to GA led to studies in animal models of various cancers. However, GA suffers from several target-unrelated limitations as an in vivo chemotherapeutic agent because of its chemical structure, as it contains a benzoquinone ring, rendering it hepatotoxic [15]. GA has been extensively modified to limit these liabilities and some of the resulting derivatives are still undergoing clinical assessment (reviewed in [16]). However, most recent efforts have been directed at developing synthetic small molecule inhibitors of distinct chemical scaffold, such as the purine-scaffold

Author Summary

Adult filarial worms are long-lived nematode parasites that have proved very difficult to kill with existing drugs. Current campaigns for the control or elimination of these parasites are largely based on treatment with drugs such as diethylcarbamazine or ivermectin, that preferentially kill the first stage larvae of the parasite, the microfilariae. As microfilariae repopulate the body from unaffected adult worms, repeated dosing with these drugs is required over the long reproductive life span of the worm. The availability of compounds with macrofilaricidal activity would help facilitate the goal of controlling filarial infections. Hsp90 is a recognized target in tumor cells: consequently many oncology programs have developed small molecule inhibitors of Hsp90, several of which are commercially available. Here we provide proof of principle that inhibition of Hsp90 is lethal to adult Brugia worms in vivo, as well as in vitro, suggesting that these compounds may have potential for further development as macrofilaricidal drugs.

series [17], that bind at the same site as GA but lack the targetunrelated liabilities. These molecules have undergone considerable modification and one compound, PU-H71, shows potential in the clinic [18,19]. Several additional N-terminal Hsp90 inhibitors have been identified in high throughput screens, including the pyrazole, isoxazole and triazole resorcinol classes such as VER-50589, NVP-AUY922 and STA-9090 (ganetespib), respectively [20,21]. NVP-AUY922 is progressing through Phase I/II clinical trials while STA-9090 has advanced to Phase III [22,23]. An additional class of compound, the Serenex series, also progressed to phase I/II clinical trials (reviewed in [22]). In this paper we report the efficacy of five inhibitors, representing four different classes of compound, on adult Brugia in vitro and compare the results with those from screens based on Mf viability and a fluorescence polarization assay. We focus on the most active compound, NVP-AUY922, and describe its in vitro effects on three life cycle stages of Brugia and its efficacy against adult worms in vivo.

Methods

Ethics statement

All animal protocols were carried out in accordance with the guidelines of the UK Home Office, under the Animal (Scientific Procedures) Act 1986, following approval by the University of Glasgow Ethical Review Panel. Experiments were performed under the authority of the UK Home Office, project numbers 60/4448 and 60/3792.

Parasites

The *Brugia pahangi* life cycle was maintained by serial passage through mosquitoes (*Aedes aegypti*, Refm) and jirds, *Meriones unguiculatus*, as described previously [24].

Preparation of worm extracts

Adult worms of *B. pahangi* were obtained from infected jirds after 3–4 months, exactly as described previously [11] and were frozen in liquid nitrogen, ground in a pestle and mortar to a fine powder and re-suspended in an appropriate volume of HFB assay buffer (20 mM HEPES, pH 7.3, 50 mM KCl, 5 mM MgCl₂, 20 mM Na₂MoO₄, 1% NP40). Protein concentrations were estimated using the BioRad protein assay. At this point lysates were freezedried for shipping to the USA.

Fluorescence polarization assay

The FP assay was set up essentially as described previously [12,25]. In brief, assays were performed in black 96-well halfvolume non-binding microtiter plates (Corning #3686) in a total volume of 50 µl per well. Assay buffer (HFB2) contained 20 mM HEPES, pH 7.3, 50 mM KCl, 2 mM EDTA, 0.01% Triton-X100, 0.1 mg/ml bovine gamma globulin (Sigma #G5009, Saint Louis, MO), 2 mM DTT (Sigma, Saint Louis, MO) and protease inhibitor cocktail (Roche #11836170, Indianapolis, IN). The equilibrium binding of Cy3B-GA and recombinant human Hsp90a (Enzo Life Sciences, Farmingdale, NY USA) or parasite lysate was determined by creating a two-fold dilution series of protein/extract for an eleven-point curve with the first column containing no protein. The dilution series was incubated with 6 nM Cy3B-GA in triplicate at 4°C with gentle shaking for different periods of time and FP measurements taken on a Safire2 plate reader (Tecan, San Jose, CA) with excitation and emission wavelengths of 530 nm and 585 nm, respectively, and a bandwidth of 20 nm. All FP values are expressed in millipolarization (mP) units with the mP value of free Cy3B-GA probe set to 50. Equilibrium binding constants were determined by nonlinear regression using a one-site binding model (GraphPad Prism software).

Competition FP assays

The relative binding affinities of inhibitors to human or parasite-derived Hsp90 was determined using competitive FP binding assays. Human Hsp90 was used at a concentration resulting in 50% maximal binding of 6 nM Cy3B-GA (2.4 nM for human Hsp90a). For parasite-derived Hsp90, an amount of parasite lysate resulting in 50% of maximal Cy3B-GA binding was used. The drugs tested in the FP assay were GA and 17-AAG, CCTC018159, VER-49009, VER-50589, NBP-AUY922, NVP-BEP800, CAY 10607, BIIB021, PU-H71, SNX-2112, SNX-9203 and HSP990. Stock solutions of each compound were prepared in DMSO at a concentration of 10 mM and 3-fold serial dilutions prepared in DMSO for eleven point curves. Drugs were then diluted 100-fold into HFB2 assay buffer containing 12 nM Cy3B-GA in 96-well storage plates to create 2X drug solutions. Drug solutions (25 μ l/well) were then transferred in duplicate to 96-well black assay plates (Corning#3686) containing 25 µl HFB2 with 2X the final desired concentration of Hsp90. The final concentration of Cy3B-GA was 6 nM and the final DMSO concentration in all wells was 0.5%. Free Cy3B-GA (mP set to 50) and buffer only (background) wells were included as controls on each plate. Plates were incubated at 8°C with gentle shaking for 20 h. FP measurements were taken and the inhibitor concentration at which 50% of bound Cy3b-GA was displaced (IC₅₀) was determined using nonlinear regression with a four parameter logistic equation (GraphPad Prism software).

Drugs

The five new compounds selected for *in vitro* testing on *B. pahangi* were NVP-AUY922, NVP-BEP800, SNX-2112, SNX-9203 and BIIB021. GA was used as a positive control in some experiments. All compounds were supplied by Selleck Chemicals (www. selleckchem.com), with the exception of GA, which was supplied by MBL International Corporation (Woburn, MA). Drugs were dissolved in DMSO to give a stock solution of 10 mM, then aliquoted and stored at -20° C. Working concentrations of drugs were prepared on the day of use by dilution in tissue culture medium. For *in vivo* studies, NVP-AUY922 was purchased from LC Laboratories (www.LCLabs.com) and dissolved in DMSO at 50 mg/ml.

Effect of inhibitors on B. pahangi viability in vitro

As adult Brugia worms are limited in numbers, initial experiments assessed the effect of each drug on Mf viability. Mf were purified from infected animals essentially as described previously [26]. In brief, following lavage of the peritoneal cavity of an infected animal with Hanks Balanced Salt Solution (HBSS) prewarmed to 37°C, Mf were collected by centrifugation and then purified from host cells by centrifugation through lymphoprep (Sigma). This procedure was repeated twice. Mf were collected from the pellet, washed twice in HBSS and once in worm culture medium (WCM) which comprised RPMI 1640, (Invitrogen Cat No: 52400), containing 5% heat inactivated fetal calf serum, 1% glucose, 100 units/ml penicillin and 100 ug/ml streptomycin (all Invitrogen). Mf were then dispensed into the wells of a 24-well plate to give approximately 200 Mf in 2.0 ml, using a single well for each drug concentration. All procedures were carried out using aseptic techniques. The five novel compounds, plus GA and medium alone controls, were tested three times against Mf over the full range of concentrations.

For adult worm assays, adult female B. pahangi, 3-4 months old, were incubated individually in 2.0 ml of WCM overnight in 24well plates and pre-screened for Mf production. Any worms that failed to produce Mf overnight were discarded. For the drug experiments, six female worms of B. pahangi for each concentration of drug were cultured individually in 24-well plates in 2.0 ml of WCM containing drug, or carrier alone (DMSO) at a concentration equal to that in the highest concentration of drug. In some experiments, GA was used as a positive control. Initially, all five compounds were tested over selected concentrations starting at 2 µM to 100 nM (see Results for details). For Hsp90 inhibitors, Mf output by individual female worms is a sensitive indicator of adult worm viability and, in most experiments, was assessed at 72 h. In addition, adult worms were examined microscopically on a daily basis for 7-10 days to determine whether lower concentrations of drugs had any effects over a longer period of incubation. Results are expressed as mean Mf output ± SD over a 72 h period. Statistical significance between groups was calculated using the Mann Whitney test with P values < 0.05 being considered significant.

In two additional experiments, adult worms were exposed to a short 24 h incubation in 250, 25 or 10 nM NVP-AUY922 or DMSO in medium alone, using six worms per concentration as described above. After 24 h in drug, adult worms were removed, washed out of drug and incubated in medium alone. Mf output was counted after 24 h in drug and again after 24 h or 48 h in medium alone. Plates were maintained for up to 10 days and the condition of adult parasites noted at regular intervals. In one experiment the effect of GA on Mf output by adult B. malayi worms was compared with B. pahangi. B. malayi worms were kindly provided by Prof. R. Maizels (University of Edinburgh). In this experiment, Mf were counted after 48 h of culture in GA at 2.0 µM and 1.0 µM GA. In all in vitro experiments, plates were viewed daily and the motility and condition of the parasites noted by two independent observers, of whom one was unaware of the contents of the well.

L3 stages were harvested from mosquitoes nine days postinfection. L3 were picked individually with a fine glass pipette, counted and washed three times in HBSS containing 1000 units/ ml of penicillin and 1000 μ g/ml streptomycin by sedimentation at room temperature. 20–30 L3 per well were plated out in duplicate in 24-well plates in 2.0 ml of WCM containing drug, or carrier alone (DMSO) and cultured at 37°C for up to 7 days. NVP-AUY922 was tested at a range of concentrations: 500, 250, 100, 50, 25, 10, 5.0, 1.0, 0.5 and 0.1 nM. This experiment was repeated twice.

In vivo testing of NVP-AUY922

Adult worms were removed from the peritoneal cavity of infected jirds and rinsed in HBSS. 10 adult female worms were transplanted into the peritoneal cavity of each of ten male BALB/c mouse using standard methods [27,28]. Adult worms were transferred into the peritoneal cavity of anaesthetized mice using a blunted glass hook. Not all mice received the full quota of 10 worms (see Table 1). The stock solution of drug was diluted to 5 mg/ml in sterile PBS containing Tween 20, to a final concentration 5%, and DMSO to a final concentration of 10%, as detailed previously [29]. Five mice were treated with 50 mg/Kg NVP-AUY922 and five with sterile PBS/Tween 20/DMSO by intra-peritoneal injection at three time points: day 0 (7 days posttransplantation), day 3 and day 7. This dose was selected on the basis of previous studies in a mouse xenograft model [29]. Adult worms and Mf were recovered 9 days after the last dose of drug by peritoneal lavage with pre-warmed HBSS. The condition of any adult worms recovered was noted. Mf in the first 12 ml of peritoneal washings were pelleted by centrifugation and fixed in 2% formalin in water and stored at 4°C until counted. Mice were weighed at each time point to monitor any possible weight loss.

Results

A fluorescence polarization (FP) assay shows binding of various classes of Hsp90 inhibitor to both *B. malayi* and *B. pahangi* and to human Hsp90 α

Several classes of inhibitor were screened against *B. pahangi or B.* malayi lysates, as well as human Hsp90a, using the FP assay originally developed as a high throughput screen for Hsp90 inhibitors in tumor cells, as previously applied to Brugia [12]. This assay is based on the ability of small molecules to inhibit the binding of Cy3B labeled GA to Hsp90. Table 2 shows the range of IC₅₀ values for the binding of selected compounds to *B. pahangi*, *B.* malayi or human Hsp90a. In general, most compounds bound to parasite-derived and human Hsp90 with broadly similar affinities. Several compounds bound worm Hsp90 with high affinity, including NVP-AUY922 and VER-50589, each of which bound Brugia Hsp90 at low nanomolar concentrations (IC₅₀ 1–2 nM). A second group of compounds including GA, PU-H71, SNX-2112, SNX-9203, BIIB021 and VER-49009 bound Brugia Hsp90 with an IC₅₀ in the range of 10-20 nM. Most drugs bound to Hsp90 from both species of Brugia with broadly similar affinity; CAY10607 was

Table 1. Recovery of adult worms and Mf from mice post-NVP-AUY922 treatment.

Group	Treated	Control
No. mice with worms	5/5	5/5
No. mice with viable worms	2/5	5/5
Total number worms recovered	8/48**	26/47**
No. viable worms recovered (mean % \pm SD)	2 (4.4±6.0)	24 (51.6±18.5)
No. mice with live Mf	2/5	5/5
Mean no Mf per mouse	4730±5101	45,880±17687

**not every animal received the full complement of 10 female worms. Recoveries from individual drug-treated mice: 3/10; 1/10; 2/9; 1/9; 1/10. Only 2 of the 8 worms recovered were normal, all others were coated in cells

and largely immobile Recoveries from individual control animals: 8/9; 3/10; 4/10; 6/10; 5/10. All worms recovered from control animals showed normal motility and had no adherent

cells with the exception of a single worm, in each of two mice.

doi:10.1371/journal.pntd.0002699.t001

an exception in this respect, showing a 10-fold higher affinity for *B. pahangi* Hsp90 compared to *B. malayi* Hsp90.

Pre-screening using Mf identifies a spectrum of drug activity

We initially compared the efficacy of Hsp90 inhibitors against Mf as this life cycle stage is extremely abundant, while the numbers of adult worms are more restricted. In these experiments six drugs, representing five different chemotypes were selected including NVP-AUY922, NVP-BEP800, SNX-2112, SNX-9203, BIIB021 and GA (see Fig. 1 for drug structures), were tested over ten doubling dilutions from 4.0 μ M to 0.976 nM and compared with the DMSO vehicle. The results of these experiments were clear-cut: NVP-AUY922 was by far the most effective compound tested, killing 100% of Mf by day 7 at all concentrations down to and including 1.95 nM. At the lowest concentration of NVP-AUY922 tested (0.976 nM), \sim 75% of the Mf were dead after 7 days incubation. In contrast, a concentration of 500 nM NVP-BEP800 was required to kill the majority of Mf by day 7, with \sim 50% death at 250 nM. Lower concentrations of NVP-BEP800 affected Mf motility but did not kill substantial numbers of worms. For the SNX compounds, SNX-2112 killed approximately 50% of Mf at 500 nM by day 7, while at 250 nM, Mf were very sluggish but still alive. SNX-9203 was slightly more effective at lower concentrations, killing ~90% of Mf at 500 nM after 7 days exposure, while at 250 nM most worms were alive but were much less motile than controls. Finally, BIIB021 was active only at 4.0 µM, while lower concentrations affected worm motility but did not kill the worms within a 7-day period. In comparison, GA killed \sim 90% of the Mf at 31.25 nM, while at 15.6 nM \sim 50% of the Mf died, similar to the effective dose reported previously [11]. Thus, NVP-AUY922 was the most efficient inhibitor of Hsp90 as judged by Mf killing followed by GA, while BIIB021 was the least effective and the SNX compounds and NVP-BEP800 had broadly similar effects on Mf viability (see Table 3 for summary).

Table 2. Binding affinity of various Hsp90 inhibitors versus human Hsp90α, *B. pahangi* and *B. malay*i Hsp90 in the FP assay.

Compound	HsHsp90α IC ₅₀ (nM) ^ª	<i>B. pahangi</i> IC ₅₀ (nM) ^a	<i>B. malayi</i> IC ₅₀ (nM)ª
Geldanamycin	8.7 (7.1–11)	6.4 (5.3–7.8)	12 (9.5–15)
17-AAG	12 (8.6–18)	8.0 (7.1–9.0)	2.5 (2.0–3.3)
CCT018159	270 (190–400)	240 (180–320)	330 (280–390)
VER-49009	8.0 (5.8–11)	11 (8.1–15)	14 (11–17)
VER-50589	1.5 (1.3–1.8)	0.54 (0.40–0.72)	1.2 (0.99–1.4)
NVP-AUY922	1.1 (1.0–1.4))	0.64 (0.42–1.0)	0.95 (0.68–1.3)
CAY10607	2.2 (1.7–2.8)	1.6 (1.3–2.0)	16 (12–22)
NVP-BEP800	13 (8.7–18)	6.9 (3.7–13)	8.7 (6.2–12)
BIIB021	2.6 (1.7–4.2)	16 (14–18)	14 (12–17)
PU-H71	10 (6.6–15)	17 (14–21)	27 (22–34)
SNX-2112	4.6 (3.5–6.0)	15 (6.7–35)	20 (13–33)
SNX-9203	3.8 (2.8–5.0)	10 (5.3–19)	11 (7.1–18)
HSP990	1.4 (1.1–1.8)	nd	2.3 (1.6–3.3)

^aBest fit (95% confidence intervals).

nd = not determined.

doi:10.1371/journal.pntd.0002699.t002

NVP-AUY922 affects Mf output by adult worms at low nanomolar concentrations

We have previously described a sensitive assay to record the effect of GA and the purine scaffold inhibitors on adult worm viability that involves counting Mf output over a designated time of exposure to inhibitor [11,12]. In initial experiments, all five new compounds were screened against adult female worms at concentrations ranging from 2.0 µM to 100 nM and Mf output and adult worm viability monitored over a 7-day period (data not shown). Three of the five compounds tested (NVP-AUY922, NVP-BEP800 and SNX-9203) had a significant effect on Mf output at all doses tested including 100 nM, while SNX-2112 was significant only at 500 nM and BIIB021 at 2.0 µM. In additional experiments, NVP-BEP800 and the SNX compounds were further titrated from 500 nM to 1.0 nM to estimate the minimal effective concentration. For all three drugs a concentration of 500 nM was required to consistently show a significant effect on Mf output by adult worms in replicate experiments.

Further experiments dealt only with the most effective compound NVP-AUY922, which was titrated over a range of concentrations from 500 nM to 1.0 nM. Following 72 h of exposure to all concentrations from 500 nM to 10 nM NVP-AUY922, a significant inhibitory effect on Mf output was observed (P = 0.0087 for 10 nM vs DMSO, see Fig. 2). Continuing the cultures for a further three days in the presence of drug resulted in a significant decrease in Mf output at 5.0 nM (P=0.0043) but not at 1.0 nM. By 6 days of exposure all adult worms were dead at concentrations of 100 nM NVP-AUY922 and above, and, in a typical experiment (shown in Fig. 2), 4/6 worms were dead at 50 nM drug and 3/6 were dead at 25 nM drug. The remaining worms although alive, were extremely sluggish. Although the motility of the adult worms was affected by exposure to lower concentrations of NVP-AUY922 (10 and 5.0 nM), they did not die at these concentrations. Worms that were dving tended to burst and release their uterine contents and wells contained many embryonic stages in addition to Mf. Confirmation that NVP-AUY922 has a direct macrofilaricidal effect was obtained by exposing male worms to a range of concentrations from 5.0 µM to 100 nM. By day 7 of exposure 100% of male worms exposed to 100 nM drug were dead.

NVP-AUY922 affects the L3 stage of B. pahangi

While for human filarial parasites, drugs that target adult worms are the priority, it was of interest to determine whether NVP-AUY922 also killed the infective form of the parasite, the L3. In these experiments, L3 were harvested directly from mosquitoes, washed and exposed to varying concentrations of NVP-AUY922 from 500 nM to 0.1 nM in WCM. After 6 days exposure to drug, 100% of L3 were dead at all concentration down to and including 10 nM. At 5.0 nM approximately 30% of parasites were dead and the remainder were moving very slowly. At this time point there was no significant mortality in control wells and, at concentrations of 1 nM or below, no effect on the L3 was observed. Thus NVP-AUY922 is toxic to L3 stages at relatively low concentrations.

NVP-AUY922 is effective against adult worms following a short-term exposure *in vitro*

In the experiments described above, adult worms or Mf were continuously exposed to drug and viability assessed. However, as any drug that might be used against filarial worms *in vivo* would be required to exert its effect over a limited period of time, we carried out two additional experiments to determine the outcome of exposing adult female worms to NVP-AUY922 for a 24 h period only. In these experiments six adult female worms per group were



Figure 1. Structures of Hsp90 inhibitors used in the *in vitro* experiments. Figure shows chemical structures of Geldanamycin (GA), NVP-AUY0922, NVP-BEP800, SNX-9203 and SNX-2112. doi:10.1371/journal.pntd.0002699.g001

cultured individually with 250, 25 or 10 nM NVP-AUY922 or the appropriate concentration of DMSO alone for 24 h. Worms were then washed free of drug and cultured in medium alone for a further 7-9 days. After 24 h in drug, significantly fewer Mf were produced at 250 nM and at 25 nM NVP-AUY922 (P = 0.0043 for both concentrations) but no significant difference was observed in Mf output in worms incubated with 10 nM drug. After 24 h in drug and 24 h in medium alone, Mf production was almost completely inhibited from worms cultured in 250 nM drug (mean of 11±23 Mf) or 25 nM drug (mean of 100±60 Mf) compared to DMSO controls (1658±198, P<0.05 for both 250 nm and 25 nM drug). Although worms exposed to 10 nM drug for 24 h produced fewer Mf (mean 910±674 Mf) than DMSO controls this difference failed to reach significance (P = 0.0635). However following 24 h in drug and 48 h in medium alone, adult worms exposed to 10 nM NVP-AUY922 for 24 h produced significantly fewer Mf than control worms (P = 0.0317) (see Fig. 3). Adult worms were clearly affected by a short-term exposure to 250 nM NVP-AUY922, being much less motile than control worms after 24 h in

Table 3. Minimal effective doses of various compounds onMf viability.

Drug	Concentration	% death by day 7
NVP-AUY922	1.95 nM	100%
	0.976 nM	75%
NVP-BEP800	250 nM	50%
SNX-9203	500 nM	90%
SNX-2112	500 nM	50%
BIIB021	4.0 μM	100%
GA	31.25 nM	90%
	15.6 nM	50%

Mf were cultured in 24-well plates in 2.0 ml medium containing drug at doubling dilutions from 4.0 μ M to 0.976 nM. Plates were scored daily by microscopy and the percentage of worms alive estimated at each time point. Results are representative of three separate experiments. doi:10.1371/journal.pntd.0002699.t003

medium alone. By 48 h they were elongate and by 9 days in medium alone, they were barely moving, but they were still alive. Worms exposed to 25 nM NVP-AUY922 were also noticeably more sluggish than control worms at day 9, while no obvious difference was observed between those incubated in 10 nM drug and DMSO controls. The experiment was discontinued at this point.

B. malayi and *B. pahangi* adult females are equally sensitive to inhibition of Hsp90

Most of our experiments have been carried out using *B. pahangi*, a close relative of the human parasite, *B. malayi*. We compared the efficacy of GA at two concentrations (1.0 and 2.0 μ M) on *B. malayi* in parallel experiments with *B. pahangi*. Not surprisingly, given the degree of amino acid identity between Hsp90 in the two species (93.5% identical), both were equally sensitive to Hsp90 inhibition. In this experiment, the reduction in Mf output at any one concentration of drug was almost identical: at 2.0 μ M GA, there was a 93% reduction in Mf output in *B. malayi* and 92% with *B. pahangi* while at 1.0 μ M, there was a 74% reduction in Mf output with *B. malayi* and a 78% reduction with *B. pahangi* after 48 h of drug exposure (P=0.0043 for all concentrations of GA versus DMSO control).

Testing NVP-AUY922 in vivo

As NVP-AUY922 appeared to be extremely active at low concentrations after a short exposure in vitro, it was pertinent to determine whether it would have activity in vivo against adult worms transplanted into the peritoneal cavity of BALB/c mice. Three animals in the treated group were given 10 adult worms, while the remaining two mice received 9 worms, while in the control group, one animal received 9 worms, one received 8 worms and the remaining three mice received 10 worms. Following adult worm transplant, mice were randomly assigned to a treatment group (five per group) and received either 50 mg/ Kg NVP-AUY922 at three time points by intra-peritoneal injection or an injection of PBS/Tween 20/DMSO. Mice were weighed at each treatment and prior to recovery of adult worms, but no weight loss was observed in drug-treated animals over the time course of the experiment. Adult worms were recovered 9 days after the last injection of drug, at which point there was a significant reduction in worm recovery from treated mice compared to control animals (P = 0.0109) (see Table 1). All control animals contained live, motile adult worms with recoveries varying from 30-78% of transplanted worms (see Table 1 for details). In contrast, very few live worms were recovered from NVP-AUY922 treated mice (recovery of live worms ranged from 0–11%, Table 1). Adult parasites recovered from all animals were placed in HBSS at 37°C for 2-3 hours and examined again. There was no evidence to suggest that the adult worms recovered from drug-treated animals regained their motility over this time period. In three out of five treated animals, only a few dead Mf were observed in the peritoneal washings, while the remaining two animals contained low numbers of Mf that were very slow moving (P = 0.0079, NVP-AUY922 versus control).

Discussion

In this paper we extend our observations on the effect of Hsp90 inhibitors on adult *Brugia* worms *in vitro* and *in vivo*. A panel of commercially available Hsp90 inhibitors, all designed and optimized for binding to human Hsp90, was profiled in FP binding assays. The broadly similar binding affinities between human and parasite proteins highlight the evolutionarily



Conc NVP-AUY922 in nM

Figure 2. NVP-AUY922 significantly reduces Mf output from female worms at 10 nM. Graph shows mean (\pm SD) Mf output from groups of six female *B. pahangi* worms after 72 h exposure to NVP-AUY922 at 500, 100, 50, 25 10, 5 and 1 nM compared to DMSO control. Mf output is significantly different from control at all concentrations to 10 nM NVP-AUY922 (P \leq 0.0087). Results shown are representative of three experiments. In this experiment, cultures were continued for a further 3 days and Mf counted again. At this time point (a total of 6 days exposure to drug), Mf output was significantly different from controls at 5 nM NVP-AUY922 (P=0.0043) but not at 1 nM. doi:10.1371/journal.pntd.0002699.q002

conserved structure of the nucleotide-binding domain that is targeted by these inhibitors. Selectivity towards parasite Hsp90 would obviously be preferable, but this endeavor would require a structure-based design effort, as recently described for trypanosome Hsp83 (Hsp90) [30]. In the spirit of exploring the potential for a direct repurposing strategy, the translation from parasite Hsp90 binding to filaricidal activity was examined. Five clinically viable compounds belonging to four different drug classes were tested *in vitro* for their ability to kill Mf and inhibit Mf output from adult female worms. The results from all three systems were reasonably consistent and highlight the efficacy of one specific inhibitor, NVP-AUY922. This compound showed significant activity against adult female worms at a concentration of 25 nM after 6 days exposure, significantly inhibiting Mf output and killing 50% of adult worms. Although lower concentrations of NVP-AUY922 (down to 5.0 nM) showed a significant effect on Mf



Figure 3. A 24 h exposure to NVP-AUY922 reduces Mf output from female worms. Graph shows the mean (\pm SD) Mf output from groups of six female *B. pahangi* worms exposed to 250 nM, 25 nM or 10 nM NVP-AUY922 for a period of 24 h only (time 0). Worms were washed free of drug and cultures continued in medium alone for a further 24 h or 48 h and Mf output quantified. P<0.05 for 250 nM and 25 nM compared to control at all time points; for 10 nM drug, P<0.05 for 48 h only. Results shown are representative of two experiments. doi:10.1371/journal.pntd.0002699.q003

output, adult worms were not killed at this concentration over the time scale of the experiment. NVP-AUY922 showed a high affinity for Brugia Hsp90 in the FP screen, further validating the application of this assay as a high throughput screen. It was also the most active compound tested in the Mf killing assay. As a single infected animal can produce millions of Mf, these results indicate that using Mf could provide a simple, inexpensive and relatively high throughput pre-screen for compounds with macrofilaricidal activity, while acknowledging that not all compounds that effect Mf will also kill adult worms. The translation from Brugia Hsp90 binding to micro- and macro-filaricidal activity depends upon the chemotype being tested and this is evident in the compounds used here. While NVP-AUY922 bound Brugia Hsp90 at ~ 1 nM and had an EC₅₀ against Mf viability in the same range, the other compounds exhibited variable translation between the two assays. For example, NVP-BEP800 bound quite well to Brugia Hsp90 (\sim 7 nM IC₅₀) but required much higher concentrations to kill microfilaria (~250 nM EC₅₀) during an extended 7-day incubation. There are many possible explanations for this discrepancy, including failure of the compound to fully equilibrate within the microfilariae during the experiment. It is possible that this particular compound does not penetrate well or is actively exported from the parasites. As more detailed studies would be required to address this issue, we continued to focus on compounds with the best activity against live parasites.

We assessed the effect of these inhibitors on adult female worms by quantifying Mf output, as a measure of worm viability. Mf output is an active process in filarial worms and relies on the presence of live Mf in utero and on the activity of the vulva, a muscular opening close to the anterior end of the worm. When Hsp90 is inhibited, cessation of Mf output is a sensitive surrogate measure of adult female worm health and appears to be one of the first signs that worm viability is compromised. Once Mf production ceases following exposure to Hsp90 inhibitors, it does not resume, at least over the time scale of our in vitro experiments. In addition, previous studies [11] demonstrated that embryogenesis in adult *B. pahangi* was disrupted upon inhibition of Hsp90 with GA. When RNAi was used to knockdown hsp90 (daf-21) in the free-living nematode Caenorhabditis elegans, one of the most penetrant phenotypes observed was a protruding vulva and sterility in the F1 generation [31]. In C. elegans, the vulva is a complex structure, the development of which is dependent upon a number of signaling pathways (reviewed in [32]). The vulva is innervated by neurons and egg-laying in C. elegans (the equivalent of Mf release in filarial worms) is regulated by multiple factors and molecules, including neuropeptides, kinases and members of the TGF- β signaling pathway [33].

Much less is known of the factors that regulate Mf production in filarial nematodes, but the vulva may be particularly sensitive to Hsp90 inhibitors, perhaps acting on various kinases required for signaling in this structure. Alternatively, the rapid inhibition of Mf output observed (within 24 h at high concentrations of inhibitor) may reflect a general demise in adult female worms upon inhibition of Hsp90. In the related filarial nematode O. volvulus, ivermectin is reported to inhibit Mf output, while not killing the adult worm. In cattle infected with O. ochengi, embryogenesis is disrupted by treatment with ivermectin and an accumulation of dead and dying Mf are observed [34], similar to that described for female Brugia exposed to GA [11]. However, the results of the present study show that inhibition of Hsp90 by NVP-AUY922 not only affects Mf release but also directly affects the viability of adult worms. This property was clearly demonstrated following in vitro exposure of both male and female adults to the inhibitor and in vivo administration of the inhibitor to mice following transplant of female worms. Hsp90 functions in a complex with other proteins to fold and/or stabilize a wide variety of client proteins, most of which have been identified from mammalian cells or yeast. A list of Hsp90 client proteins is curated by the Picard Lab (http://www.picard.ch). However, we know little about the key client proteins clients in *Brugia*, except by homology with known interacting proteins identified from other systems. A better understanding of the Hsp90 interactome in *Brugia* may help explain the effects observed upon chemical inhibition of Hsp90.

As all our previous studies have focused on *B. pahangi*, we also tested *B. malayi* in this study and showed that GA had very similar effects upon both species. Similar data was recently published [35] showing that GA and four derivatives of GA, were active *in vitro* on adult *B. malayi* and the trematode parasite *Schistosoma japonicum*. In that study, the lowest concentration tested was 500 nM, but encouragingly all compounds were active against adult *B. malayi* at this concentration.

Interestingly, NVP-AUY922 also showed potent activity against the L3 stage of *B. pahangi*. While drugs that target the L3 stage of filarial nematodes are not the highest priority for human use, they are of significant interest in the veterinary field, where prophylaxis of the dog heartworm, *Dirofilaria immitis*, is a major area of concern in veterinary practice. Here, the macrocyclic lactones, such as ivermectin and moxidectin, have been the mainstay of control for many years. Given the recent observation on treatment failures with ivermectin in some dogs in the Southern states of the USA [36], there is a clear need for the development of novel compounds to protect susceptible animals.

As a preliminary to in vivo testing, we investigated whether a short exposure to the most effective compound identified, NVP-AUY922, affected adult worm viability in vitro. These experiments showed that Mf output was significantly reduced following a 24 h exposure to all concentrations of drug including 10 nM (the lowest concentration tested in this study), although it required an additional 48 h incubation in medium alone to detect this effect. However, although Mf output was significantly different between drug exposed and control worms, a direct effect was observed at 250 nM concentration after a 24 h exposure, with adult female worms being almost motionless after 9 days in culture. While a 24 h exposure to 25 nM NVP-AUY922 resulted in a significant decrease in motility, the adult female worms were still alive at day 9. Whether such motility-impaired worms would be viable in vivo in an immuno-competent host is doubtful. These data also highlight one of the characteristics of Hsp90 inhibitors: the time taken to kill adult worms. They do not kill rapidly, which may be a useful attribute for in vivo use, as some of the pathogenesis of lymphatic filariasis is associated with death of the adult worms, and the subsequent release of antigen and the Wolbachia endosymbiont [37]

NVP-AUY922 belongs to the isoxazole resorcinol class of Hsp90 inhibitors and has shown some promise as an anti-tumor agent in a mouse xenograft tumor model [29,38]. In the first of these studies, pharmacokinetic analysis showed that the plasma levels of drug reached a maximum of 52,506 nM at 0.25 h post-dosing following a single dose of 50 mg/Kg delivered by the intra-peritoneal route. As a 24 h exposure to 250 nM NVP-AUY922 had a significant deleterious effect on adult female worms, we were encouraged to test this compound *in vivo*, using an adoptive transplant system in which viable adult worms are transplanted into the peritoneal cavity of mice [27,28]. In previous experiments with NVP-AUY922 administered in a similar dosing regimen to nude mice containing xeno-grafted tumors, weight loss was observed [38]. However our immune-competent animals showed no loss in weight or other obvious deleterious effects over the time scale of the experiment.

Adult worms were recovered nine days after the last dose of drug, as the in vitro data indicated that it might take some time for the drug to act. Additionally, previous studies with suramin, one of the few compounds with activity against adult filarial worms, demonstrated that it took approximately 6 weeks to reduce adult worm recovery in the jird model [39]. A total of eight out of 48 transplanted female worms were recovered from drug treated animals, but of these only two were healthy with the others being coated in cells, indicating that they were in the process of being cleared by the immune system. In contrast, 26 female worms out xof 47 transplanted were recovered from animal treated with vehicle alone. Moreover, only two of the drug-treated mice contained live Mf, and these appeared very sluggish compared to controls. Further studies will be required to examine alternative routes of drug administration and to explore in more detail the mechanism by which these inhibitors exert their macrofilaricidal effect.

The clinically viable Hsp90 inhibitors tested here are targeted and potent towards human Hsp90 and thus may have unwanted side effects. In this respect, the therapeutic potential of any of these inhibitors for filarial disease will depend on the pharmacodynamic differences between host toxicities and parasite killing. There are pharmacokinetic data in both animals and humans available for many of the Hsp90 inhibitors. We have used this data in combination with in vitro worm killing assays to predict the doses of NVP-AUY922 that would be successful in killing parasites in vivo. While efficacy in cancer treatment likely requires chronic dosing of near maximum tolerated doses, the treatment of parasitic diseases would ideally necessitate a short course of treatment. It is not currently clear whether a short treatment course with current inhibitors would be able to clear parasites, while maintaining appropriate safety margins. Additional studies will be required to carefully delineate the therapeutic window of short treatment courses. As noted earlier, the development of parasite-selective inhibitors with reduced potential for host toxicity would be of enormous benefit and should also be pursued.

References

- Dembele B, Coulibaly YI, Dolo H, Konate S, Coulibaly SY, et al. (2010) Use of high-dose, twice-yearly albendazole and ivermectin to suppress Wuchereria bancrofti microfilarial levels. Clin Infect Dis 51: 1229–1235.
- Subramanian S, Stolk WA, Ramaiah KD, Plaisier AP, Krishnamoorthy K, et al. (2004) The dynamics of *Wuchereria bancrofti* infection: a model-based analysis of longitudinal data from Pondicherry, India. Parasitology 128: 467–482.
- Basanez MG, Ricardez-Esquinca J (2001) Models for the population biology and control of human onchocerciasis. Trends Parasitol 17: 430–438.
- Bourguinat C, Ardelli BF, Pion SD, Kamgno J, Gardon J, et al. (2008) Pglycoprotein-like protein, a possible genetic marker for ivermectin resistance selection in *Onchocerca volvulus*. Mol Biochem Parasitol 158: 101–111.
- Geary TG (2012) Are new anthelmintics needed to eliminate human helminthiases? Curr Opin Infect Dis 25: 709–717.
- Neckers L (2007) Heat shock protein 90: the cancer chaperone. J Biosci 32: 517– 530.
- Shahinas D, Liang M, Datti A, Pillai DR (2010) A repurposing strategy identifies novel synergistic inhibitors of *Plasmodium falciparum* heat shock protein 90. J Med Chem 53: 3552–3557.
- Kumar R, Musiyenko A, Barik S (2003) The heat shock protein 90 of Plasmodium falciparum and antimalarial activity of its inhibitor, geldanamycin. Malar J 2: 30.
- Pallavi R, Roy N, Nageshan RK, Talukdar P, Pavithra SR, et al. (2010) Heat shock protein 90 as a drug target against protozoan infections: biochemical characterization of HSP90 from *Plasmodium falciparum* and *Trypanosoma evansi* and evaluation of its inhibitor as a candidate drug. J Biol Chem 285: 37964–37975.
- Petersen AL, Guedes CE, Versoza CL, Lima JG, de Freitas LA, et al. (2012) 17-AAG kills intracellular *Leishmania amazonensis* while reducing inflammatory responses in infected macrophages. PLoS One 7: e49496.
- Devaney E, O'Neill K, Harnett W, Whitesell L, Kinnaird JH (2005) Hsp90 is essential in the filarial nematode *Brugia pahangi*. Int J Parasitol 35: 627–636.
- Taldone T, Gillan V, Sun W, Rodina A, Patel P, et al. (2010) Assay strategies for the discovery and validation of therapeutics targeting *Brugia pahangi* Hsp90. PLoS Negl Trop Dis 4: e714.
- Robertson SA, Renslo AR (2011) Drug discovery for neglected tropical diseases at the Sandler Center. Future Med Chem 3: 1279–1288.

The ubiquitous nature of Hsp90 in normal as well as transformed cells, has led some to question the potential of this molecule as a drug target. However, for the past 20 years, the National Cancer Institute has advocated Hsp90 as a drug target, since GA was first shown to exhibit anti-tumor properties. There are currently seventeen Hsp90 inhibitors in clinical trials and a growing arsenal of novel Hsp90 inhibitors, of structurally diverse scaffold (reviewed in [40]). Current emphasis is on combination therapies in which Hsp90 inhibitors are combined with other antitumor drugs [41]. Whether a similar approach would further enhance the activity of these inhibitors against adult filarial parasites remains to be determined. However, we believe that these studies provide data to further strengthen the contention that inhibition of Hsp90 is a valid target for the chemotherapy of lymphatic filariasis, while acknowledging that a significant medicinal chemistry effort would be required to optimize the activity of these inhibitors.

Acknowledgments

We would like to acknowledge the assistance of Margaret McFadyen in helping to run the *Brugia* life cycle, the staff of Biological Services, University of Glasgow for their assistance with the worm transplant experiments, Professor R.M. Maizels, University of Edinburgh for supplying adult *B. malayi*, Dr. Marvin J. Meyers, Saint Louis University, for synthesis of the Cy3B-GA probe, Ewen Calder for assistance with ChemDraw and Drs. Collette Britton and Jean Rodgers for helpful comments on the manuscript.

Author Contributions

Conceived and designed the experiments: ED FMS VG. Performed the experiments: KO KM VG FMS ED. Analyzed the data: ED VG FMS. Contributed reagents/materials/analysis tools: ED FMS. Wrote the paper: ED VG FMS.

- 14. Pearl LH, Prodromou C (2006) Structure and mechanism of the Hsp90 molecular chaperone machinery. Annu Rev Biochem 75: 271–294.
- Taldone T, Sun W, Chiosis G (2009) Discovery and development of heat shock protein 90 inhibitors. Bioorg Med Chem 17: 2225–2235.
- Kim T, Keum G, Pae AN (2013) Discovery and development of heat shock protein 90 inhibitors as anticancer agents: a review of patented potent geldanamycin derivatives. Expert Opin Ther Pat 23: 919–943.
- Chiosis G, Lucas B, Shtil A, Huezo H, Rosen N (2002) Development of a purine-scaffold novel class of Hsp90 binders that inhibit the proliferation of cancer cells and induce the degradation of Her2 tyrosine kinase. Bioorg Med Chem 10: 3555–3564.
- Taldone T, Chiosis G (2009) Purine-scaffold Hsp90 inhibitors. Curr Top Med Chem 9: 1436–1446.
- Cerchietti LC, Lopes EC, Yang SN, Hatzi K, Bunting KL, et al. (2009) A purine scaffold Hsp90 inhibitor destabilizes BCL-6 and has specific antitumor activity in BCL-6-dependent B cell lymphomas. Nat Med 15: 1369–1376.
- Brough PA, Aherne W, Barril X, Borgognoni J, Boxall K, et al. (2008) 4,5diarylisoxazole Hsp90 chaperone inhibitors: potential therapeutic agents for the treatment of cancer. J Med Chem 51: 196–218.
- Lin TY, Bear M, Du Z, Foley KP, Ying W, et al. (2008) The novel HSP90 inhibitor STA-9090 exhibits activity against Kit-dependent and -independent malignant mast cell tumors. Exp Hematol 36: 1266–1277.
- Jhaveri K, Taldone T, Modi S, Chiosis G (2012) Advances in the clinical development of heat shock protein 90 (Hsp90) inhibitors in cancers. Biochim Biophys Acta 1823: 742–755.
- Choi HK, Lee K (2012) Recent updates on the development of ganetespib as a Hsp90 inhibitor. Arch Pharm Res 35: 1855–1859.
- Jecock RM, Devaney E (1992) Expression of small heat shock proteins by the third-stage larva of *Brugia pahangi*. Mol Biochem Parasitol 56: 219–226.
- Du Y, Moulick K, Rodina A, Aguirre J, Felts S, et al. (2007) High-throughput screening fluorescence polarization assay for tumor-specific Hsp90. J Biomol Screen 12: 915–924.
- Devaney E, Egan A, Lewis E, Warbrick EV, Jecock RM (1992) The expression of small heat shock proteins in the microfilaria of *Brugia pahangi* and their possible role in development. Mol Biochem Parasitol 56: 209–217.

- Allen JE, Lawrence RA, Maizels RM (1996) APC from mice harbouring the filarial nematode, *Brugia malayi*, prevent cellular proliferation but not cytokine production. Int Immunol 8: 143–151.
- Devaney E, Gillan V, Wheatley I, Jenson J, O'Connor R, et al. (2002) Interleukin-4 influences the production of microfilariae in a mouse model of *Brugia infection*. Parasite Immunol 24: 29–37.
- Eccles SA, Massey A, Raynaud FI, Sharp SY, Box G, et al. (2008) NVP-AUY922: a novel heat shock protein 90 inhibitor active against xenograft tumor growth, angiogenesis, and metastasis. Cancer Res 68: 2850–2860.
- Pizarro JC, Hills T, Senisterra G, Wernimont AK, Mackenzie C, et al. (2013) Exploring the *Trypanosoma brucei* Hsp83 potential as a target for structure guided drug design. PLoS Negl Trop Dis 7: e2492.
- Gillan V, Maitland K, McCormack G, Him NA, Devaney E (2009) Functional genomics of *hsp-90* in parasitic and free-living nematodes. Int J Parasitol 39: 1071–1081.
- Felix MA, Barkoulas M (2012) Robustness and flexibility in nematode vulva development. Trends Genet 28: 185–195.
- Schafer WF (2006) Genetics of egg-laying in worms. Annu Rev Genet 40: 487– 509.

- Bronsvoort BM, Renz A, Tchakoute V, Tanya VN, Ekale D, et al. (2005) Repeated high doses of avermeetins cause prolonged sterilisation, but do not kill, *Onchoerea ochengi* adult worms in African cattle. Filaria J 4: 8.
- Wenkert D, Ramirez B, Shen Y, Kron MA (2010) In Vitro Activity of Geldanamycin Derivatives against *Schistosoma japonicum* and *Brugia malayi*. J Parasitol Res 2010: 716498.
- Bourguinat C, Keller K, Prichard RK, Geary TG (2011) Genetic polymorphism in *Dirnfilaria immitis*. Vet Parasitol 176: 368–373.
- Taylor MJ, Bandi C, Hoerauf A (2005) Wolbachia bacterial endosymbionts of filarial nematodes. Adv Parasitol 60: 245–284.
- Garon EB, Finn RS, Hamidi H, Dering J, Pitts S, et al. (2013) The HSP90 Inhibitor NVP-AUY922 Potently Inhibits Non-Small Cell Lung Cancer Growth. Mol Cancer Ther 12: 890–900.
- Howells RE, Mendis AM, Bray PG (1983) The mode of action of suramin on the filarial worm *Brugia pahangi*. Parasitology 87 (Pt 1): 29–48.
- Neckers L, Workman P (2012) Hsp90 molecular chaperone inhibitors: are we there yet? Clin Cancer Res 18: 64–76.
- Lu X, Xiao L, Wang L, Ruden DM (2012) Hsp90 inhibitors and drug resistance in cancer: the potential benefits of combination therapies of Hsp90 inhibitors and other anti-cancer drugs. Biochem Pharmacol 83: 995–1004.

Copyright of PLoS Neglected Tropical Diseases is the property of Public Library of Science and its content may not be copied or emailed to multiple sites or posted to a listserv without the copyright holder's express written permission. However, users may print, download, or email articles for individual use.